Author's response to reviews

Title: Lack of association of acute-phase response proteins with hormone levels and antidepressant medication in the perimenopausal depression

Authors:

Sokratis E Karaoulanis (skar@med.uth.gr)
Katerina A Rizouli (rizoulia@otenet.gr)
Andreas A Rizoulis (andreasrizoulis@hotmail.com)
Nikiforos V Angelopoulos (nikifang@med.uth.gr)

Version: 4 Date: 17 May 2014

Author's response to reviews:

Dear Sir,

I took into account the editorial requests and I made the following changes to the final manuscript:

1. I included a title page
2. I removed the highlights section.
3. I included the abbreviations after the conclusions section.
4. I changed the wrong grammar for the sentence you noticed. The correct sentence is: Informed consent for participation in this study was obtained from each woman. The ethics committee of the University Hospital of Larissa approved this study.
5. I included a competing interests section.
6. I included an author's contributions section.
7. I included an acknowledgements section.